Author’s response to reviews

Title: Choroidal Structural Changes Correlate with Severity of Diabetic Retinopathy in Diabetes Mellitus

Authors:

Hui Wang (a502367790@126.com)
Yong Tao (502367790@qq.com)

Version: 1 Date: 08 Mar 2019

Author’s response to reviews:

Dear Editor:

Thank you for your letter. Those comments are all valuable and very helpful to us.

We have studied comments carefully and have made correction which we hope meet with approval. The main corrections in the paper are as flowing:

→ Please rename 'Materials and Methods' heading to 'Methods'.
My answer: Thank you for your kind advice. I have renamed 'Materials and Methods' heading to 'Methods' (Page 4).

→ Please reformat your Declarations section according to the submission guidelines and correct format below.
My answer: Thank you for your kind advice. I am sorry for that. I have made modifications in this part.

→ Please confirm whether the informed consent obtained from study participants was written or verbal, and clearly state this in your manuscript in the 'Ethics and consent to participate' subsection of the 'Declarations'. If verbal, please state the reason and whether the ethics committee approved this procedure.
My answer: I have added “written” in “Ethics approval and consent to participate” part and change it to “Written informed consent was obtained from each patient before the study.”
Abbreviations

My answer: I have added “List of Abbreviations” after the conclusions section.

Thank you for the kind advice.

Sincerely,

Tao Yong